Author's response to reviews

Title: DR_SEQAN: a PC/Windows-based software to evaluate drug resistance using human immunodeficiency virus type 1 genotypes

Authors:

Cesar Garriga (cgarriga@msc.es)
Luis Menendez-Arias (lmenendez@cbm.uam.es)

Version: 7 Date: 27 February 2006

Author's response to reviews:

Dear Editors,
In response to your letter of February 27th, we have introduced the requested formatting changes in the main manuscript. Thank you very much for your attention.
Sincerely,
Luis Menendez-Arias
Cesar Garriga